Author's response to reviews

Title: Abuse and disrespect in childbirth process and abortion situation in Latin America and the Caribbean - Systematic Review Protocol sofiapmadeira@gmail.com vicky.pileggi@gmail.com souzaj@who.int

Authors:

Sofia Madeira (sofiapmadeira@gmail.com)
Vicky Pileggi (vicky.pileggi@usp.br)
João Paulo de Souza (souzaj@who.int)

Version: 2 Date: 07 Feb 2017

Author’s response to reviews:

About the inclusion of Meghan's name: her name was considered to be part of the team but for various reasons she could not be part of the team. Her contribution was acknowledged in the acknowledgments section in one of the articles generated by this systematic review.

The other suggestions sent recently have been accepted and can be contemplated below.